Improving the assessment and management of obesity in UK children and adolescents: the PROMISE research programme including a RCT

Russell M Viner, 1* Sanjay Kinra, 2 Deborah Christie, 3 Tim J Cole, 4 Silvia Costa, 1 Helen Croker, 5 Tam Fry, 6 Yingfen Hsia, 7 Lee Hudson, 8 Anthony S Kessel, 9 Steve Morris, 10 Irwin Nazareth, 11 Dasha Nicholls, 8 Min Hae Park, 12 Sonia Saxena, 13 Barry Taylor, 14 Billy White 15 and Ian C Wong 16

- ¹Population, Policy and Practice Programme, Institute of Child Health, University College London, London, UK
- ²Department of Genetics and Adolescent Paediatrics, London School of Hygiene & Tropical Medicine, London, UK
- ³Department of Behavioural Science and Health, University College London Hospitals NHS Foundation Trust, London, UK
- ⁴Department of Infection, Immunology and Inflamation, Institute of Child Health, University College London, London, UK
- ⁵The Obesity and Policy Research Unit, Institute of Child Health, University College London, London, UK
- ⁶Child Growth Foundation, Edgware, UK
- ⁷Paediatric Infectious Diseases Research Group, Institute for Infection and Immunity, St George's University of London, London, UK
- ⁸Feeding and Eating Disorders Team, Great Ormond Street Hospital for Children NHS Foundation Trust, London, UK
- ⁹Director of Global Public Health, Public Health England, London, UK
- ¹⁰Centre of Applied Health Research, University College London, London, UK
- ¹¹Research Department of Primary Care and Population Science, University College London, London, UK
- ¹²Department of Health Services Research and Policy, London School of Hygiene & Tropical Medicine, London, UK
- ¹³Faculty of Medicine, Imperial College London, London, UK
- ¹⁴Paediatrics and Child Health, The Dunedin School of Medicine, University of Otago, Dunedin, New Zealand
- ¹⁵Department for Children's and Young People's Diabetes, University College London Hospitals NHS Foundation Trust, London, UK
- ¹⁶Research Department of Practice and Policy, University College London School of Pharmacy, University College London, London, UK

^{*}Corresponding author r.viner@ucl.ac.uk

Declared competing interests of authors: Stephen Morris is a member of the National Institute for Health Research (NIHR) Health Services and Delivery Research Board. Irwin Nazareth is a member of the NIHR Health Technology Assessment Commissioning Board. Sonia Saxena was funded by a NIHR Career Development fellowship (NIHR CDF-2011-04-048). Min Hae Park received grants from NIHR during the conduct of the study and received personal fees from Research Consultancy for Marie Stopes International (London, UK). Anthony S Kessel is the director of International Public Health at Public Health England.

Published March 2020 DOI: 10.3310/pgfar08030

Plain English summary

The PROMISE research programme including a RCT

Programme Grants for Applied Research 2020; Vol. 8: No. 3

DOI: 10.3310/pgfar08030

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

The childhood obesity epidemic is one of the foremost health priorities in the UK. However, there are several gaps in the childhood obesity pathway outlined by the National Institute for Health and Care Excellence. The Paediatric Research in Obesity Multi-modal Intervention and Service Evaluation (PROMISE) five-study programme was created to address these gaps.

Most parents do not recognise that their overweight/obese children are overweight, and most do not consider being overweight to be a health risk. The National Child Measurement Programme feedback was found to be helpful and had positive effects on parental knowledge and perceptions of being overweight, but had little impact on behaviours.

A small sample of general practitioners reported that using the online obesity risk prediction tool in consultations was acceptable and had potential to enhance the impact of advice. Families also found such consultations acceptable, although both doctors and families reported a need for subsequent practical support for weight management.

The multicomponent community lifestyle intervention was not more effective than enhanced standard care in reducing the body mass index (or improving any of the secondary outcomes) of obese adolescents.

There was a steady increase in anti-obesity drug and metformin (Glucophage®; Merck Serono, Darmstadt, Germany) prescribing for obese/overweight young people, but most young people rapidly stopped taking anti-obesity drugs/metformin after prescription, and several side effects were highlighted.

Deciding about weight loss surgery (bariatric surgery) for young people is a tough challenge for health-care professionals. Young people decide to undergo bariatric surgery because they 'want a different future', but the process is full of dilemmas and surgery is often chosen as 'the last resort'. Bariatric surgery in adolescence appears to be highly cost-effective.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0608-10035. The contractual start date was in January 2010. The final report began editorial review in January 2017 and was accepted for publication in March 2018. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Viner et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk